Author's response to reviews

Title: Reliability of upright posture measurements in primary school children

Authors:

Maureen P McEvoy (maureen.mcevoy@unisa.edu.au)
Karen Grimmer (karen.grimmer@unisa.edu.au)

Version: 8 Date: 28 June 2005

Author's response to reviews:

Formatting changes requested have been addressed:

1. 'Australia' has been added to the affiliation for institution 2
2. The Abstract in the submission system has been replaced so that it is identical to that in the manuscript
3. The competing interests section has been added and addressed.